Author's response to reviews

**Title:** Effects of dexmedetomidine on smooth emergence from anaesthesia in elderly patients undergoing orthopaedic surgery

**Authors:**

Dong Jun Kim (lsopp@naver.com)
Sang Hun Kim (ksh3223@chosun.ac.kr)
Keum Young So (kyso@chosun.ac.kr)
Ki Tae Jung (mdmole@chosun.ac.kr)

**Version:** 5  **Date:** 13 August 2015

**Author's response to reviews:**

Effects of dexmedetomidine on smooth emergence from anaesthesia in elderly patients undergoing orthopaedic surgery

Running title: Dexmedetomidine for emergence in the elderly

Dong Jun Kim1, Sang Hun Kim2, Keum Young So2, Ki Tae Jung2
1Department of Anesthesiology and Pain Medicine, Chosun University Hospital, Gwangju, Korea
2Department of Anesthesiology and Pain Medicine, School of Medicine, Chosun University, Gwangju, Korea

Dong Jun Kim
Department of Anesthesiology and Pain Medicine, Chosun University Hospital, 365 Pilmun-daero, Dong-gu, Gwangju 501-717, Korea

Sang Hun Kim
Department of Anesthesiology and Pain Medicine, School of Medicine, Chosun University, 365 Pilmun-daero, Dong-gu, Gwangju 501-717, Korea

Keum Young So
Department of Anesthesiology and Pain Medicine, Chosun University School of Medicine, 365 Pilmun-daero, Dong-gu, Gwangju 501-717, Korea

Correspondence to: Ki Tae Jung
Department of Anesthesiology and Pain Medicine, Chosun University School of Medicine, 365 Pilmun-daero, Dong-gu, Gwangju 501-717, Korea
Tel: 82-62-220-3223, Fax: 82-62-223-2333, E-mail: mdmole@chosun.ac.kr